Author's response to reviews

Title: Health Insurance, Neighborhood Income, and Emergency Department Usage by Utah Children 1996-1998

Authors:

Anthony Suruda (tonysuruda@aol.com)
Thomas Burns (tburns@ou.edu)
Stacey Knight (Stacey.Knight@hsc.utah.edu)
Jonathan M Dean (Mike.Dean@hsc.utah.edu)

Version: 4 Date: 9 March 2005

Author’s response to reviews:

March 9, 2005
Editor
BMC Health Services Research

Dear Ms. Shaw:

Enclosed is the final revision of the manuscript entitled Health Insurance, Neighborhood Income, and Emergency Department Usage by Utah Children 1996-98. I have made the formatting changes requested in your communication of March 8, 2005 as well as editorial changes to the text.

I thank you for your consideration.

Anthony Suruda MD MPH
tonysuruda@aol.com